

Contents lists available at ScienceDirect

Applied Soft Computing





Decision variable classification based multi-objective multifactorial memetic algorithm for multi-objective multi-task optimization problem

Zhiwei Xu^{a,b,*}, Jiafeng Xu^c, Kai Zhang^{a,b,*}, Xin Xu^{a,b}, Juanjuan He^{a,b}, Ni Wu^a

^a School of Computer Science and Technology, Wuhan University of Science and Technology, Wuhan 430065, China

^b Hubei Province Key Laboratory of Intelligent Information Processing and Real-time Industrial System, Wuhan 430065, China

^c School of Automation, China University of Geosciences, Wuhan 430074, China

HIGHLIGHTS

- A decision variables classification method based on control variable analysis is proposed.
- An evolutionary recombination strategy based on decision variables' characteristics is proposed.
- The proposed algorithm hybridizes the immune algorithm and evolutionary gradient search as the global and local search operators separately.
- The proposed algorithm can achieve better performance than the state-of-the-art evolutionary multitasking multi-objective evolutionary algorithms.

ARTICLE INFO

Keywords: Evolutionary multitasking Multi-objective evolutionary algorithm Evolutionary gradient search Decision variable classification Multi-objective multi-task optimization problem

ABSTRACT

Multi-task multi-objective optimization problems need to consider the algorithm's convergence and the population's diversity. The information transfer of decision variables with different characteristics may harm the effect of knowledge reuse. This paper proposes a novel hybrid multi-objective multifactorial memetic algorithm to address this issue. The proposed variable classification method will classify decision variables into convergence-related and diversity-related decision variables. Only the same type of decision variables in the source and target tasks can transfer information to avoid negative transfer. Different evolutionary operators are adopted according to the characteristics of decision variables during individual recombination. In addition, the proposed algorithm hybridizes the immune algorithm as the global evolutionary operator and the evolutionary gradient search algorithm as the local search operator into the multifactorial framework to enhance the searching ability. Finally, the proposed algorithm is compared with the state-of-the-art multi-objective evolutionary multitasking algorithms. The results of the experiments show that the proposed algorithm can achieve promising performance on the classical and complex multi-task multi-objective benchmark test suites.

1. Introduction

Multi-objective optimization problems (MOPs) require optimizing multiple always conflicting objectives simultaneously, which are ubiquitous in the real world [1–5]. Without loss of generality, assuming that the problem is a minimization problem, it can be defined as:

$$\min_{\mathbf{x}\in\Omega}(F(\mathbf{x}) = (f_1(\mathbf{x}), f_2(\mathbf{x}), f_3(\mathbf{x}), \dots, f_M(\mathbf{x})))$$
(1)

where $\mathbf{x}=(\mathbf{x}_1, ..., \mathbf{x}_D)$ denotes a decision vector with *D* dimension in the decision space Ω . $F(\mathbf{x}) = (f_1(\mathbf{x}), f_2(\mathbf{x}), f_3(\mathbf{x}), ..., f_M(\mathbf{x}))$ represents the objective function vector containing the *M* conflicting objectives

functions. Due to the objective functions conflict with each other, optimizing one objective function will inevitably lead to the deterioration of another objective function. It is impossible to identify one single solution optimal for all objective functions. Therefore, Pareto dominance is proposed to determine a set of best tradeoff solutions. Given two decision vector \mathbf{x} and \mathbf{y} , if $\forall i \in \{1, 2, 3..., M\}$ $f_i(\mathbf{x}) \leq f_i(\mathbf{y})$ and $\exists j \in \{1, 2, 3..., M\}$ $f_j(\mathbf{x}) < f_j(\mathbf{y})$, \mathbf{x} is said to Pareto dominate \mathbf{y} referred to as $\mathbf{x} \prec \mathbf{y}$. If there is no other solution in Ω that can dominate \mathbf{x}^* , then \mathbf{x}^* is called a Pareto optimal solutions set (PS). The objective vectors projected from the PS in the objective space are called the Pareto from (PF).

Multi-objective evolutionary algorithms (MOEAs) have been widely

https://doi.org/10.1016/j.asoc.2024.111232

Received 26 June 2023; Received in revised form 17 November 2023; Accepted 30 December 2023 Available online 4 January 2024 1568-4946/© 2024 Elsevier B.V. All rights reserved.

^{*} Corresponding authors at: School of Computer Science and Technology, Wuhan University of Science and Technology, Wuhan 430065, China. *E-mail addresses:* xuzhiwei@wust.edu.cn (Z. Xu), zhangkai@wust.edu.cn (K. Zhang).

used to solve MOPs because they can obtain multiple Pareto optimal solutions in a single run. The traditional MOEAs can be mainly classified into three categories, dominant relationship-based algorithms [6-8], performance indicator-based algorithms [9–11], and decomposition-based algorithms [12–14]. The dominance relationship-based algorithms mainly utilize the dominance relationship as the basis for screening solutions. Improving the traditional dominance relationship is the main thought for these algorithms to increase convergence pressure. The performance indicator-based algorithms usually apply the artificially designed indicator to evaluate the convergence and diversity of the current population to guide it toward the PS. The decomposition-based algorithms generally decompose a complex multi-objective problem into several single-objective problems (SOPs) or multiple simpler MOPs. This type of algorithm purports to simplify the problem, dividing and conquering. All the conventional MOEAs can usually perform decently in traditional MOPs but are designed only to solve one MOP in a single run. When faced with a new problem, the population needs to be reinitialized. However, many MOPs in the real world are interrelated. The knowledge to solve one MOP is beneficial to solve a similar one.

Inspired by the human brains that can parallelly process multiple tasks and flexibly wield the knowledge learned from different tasks, evolutionary multitasking (EMT) is proposed to optimize multiple problems simultaneously [15]. The primary purpose of EMT is to improve the optimization efficiency by transferring and reusing the knowledge learned in optimizing multiple tasks processes. The EMT is very innovative and forward-looking. Once put forward, it has aroused continuous interest and has been successfully applied to solve various real-world optimization problems [16–19]. In the EMT area, one problem that needs to be solved is called a task. The problem that optimizes multiple tasks synchronously is referred to as multi-task problem (MTOP).

When there are MOPs in the simultaneous optimization tasks, it is called a multi-task multi-objective optimization (MOMTO) problem. Unlike single-objective multitasking, multi-objective multitasking aims to find a set of nondominated tradeoff solutions for each MOP rather than a single optimal point. Therefore, the multi-objective EMT algorithms must consider not only the convergence but also the diversity of the population. The non-dominated solutions should be dispersed as much as possible in the search process. In MOPs, different decision variables have otherness characteristics, some decision variables are more related to diversity, and some decision variables are more related to convergence. The diversity-related decision variables promote solutions to evenly dispersed and the convergence-related decision variables boost the population toward the PF. During knowledge transfer, decision variables will also transfer their characteristics to the target task. When decision variables with different characteristics carry out knowledge transfer, it often adversely affects the population.

Existing multi-objective EMT algorithms often neglect this issue, and no research has made a breakthrough. To address this issue, this paper proposes a novel hybrid multi-objective multifactorial memetic algorithm (HMOMFMA). The decision variable classification method is applied to classify the decision variables into convergence-related decision variables and diversity-related decision variables according to their contribution to the convergence and diversity of the population. Specifically, only decision variables of the same category can proceed with knowledge transfer in the knowledge transfer stage. In the individual recombination stage, decision variables of different categories will be recombined using different evolutionary operators. HMOMFMA hybridizes the immune algorithm and evolutionary gradient search (EGS) into the EMT framework to enhance the search ability. The immune algorithm is utilized as the global optimization operator, which has a powerful convergence capacity by focusing on the non-dominated solutions in each iteration and can maintain the population diversity by favoring the sparse solutions selected by the maximum extension distance. The EGS is introduced as the local search operator, which has

adaptive mutation length and can guide individual evolution based on gradient information. To verify the efficacy of HMOMFMA, comprehensive empirical studies are conducted on the classical MOMTO and complex MOMTO benchmark problems. The experimental results demonstrate that the proposed HMOMFMA is superior to the state-ofthe-art multi-objective EMT algorithms.

The main contribution of this paper are summarized as follow.

1) A decision variables classification method based on control variable analysis is proposed to divide decision variables into two categories according to their contribution to the diversity and convergence of the population.

2) An evolutionary recombination strategy based on decision variables' characteristics is proposed. First, according to their characteristics, two different evolutionary operators are applied to the decision variables in the recombination process. Second, knowledge transfer occurs only between decision variables with the same characteristics.

3) The proposed HMOMFMA hybridizes the immune algorithm and EGS as the global and local search operators separately. The immune algorithm focuses on the sparsest area in the non-dominated solution set that can well maintain population diversity. The EGS performs a local search around the non-dominated solutions and can accelerate convergence by adaptively adjusting the mutation length and guiding the evolution according to the gradient information.

4) To assess the performance of the proposed HMOMFMA, experiments are conducted on the classical and complex benchmark test suites. The proposed HMOMFMA is compared with five state-of-the-art multiobjective EMT algorithms, MOMFEA [35], MOMFEA-II [26], EMT-A [32], MFEA-SADE [36], and MFEA-GHS [37] and a classic MOEA, namely NSGA-II [6]. The experimental results demonstrate that the proposed HMOMFMA is superior to other advanced EMT algorithms.

The rest of this paper is organized as follows. Section 2 reviews the related work of the proposed algorithm. Section 3 describes the details of the proposed HMOMFMA. Section 4 presents the comprehensive experiments to assess the effectiveness of HMOMFMA. Finally, Section 5 concludes this paper and prospects some future directions.

2. Related work

2.1. Multifactorial algorithm framework

Unlike SOPs and MOPs, MTOPs can be regarded as the third type of optimization problem paradigm: to optimize multiple problems simultaneously and find the optimal solution corresponding to each problem. In MTOPs, each independent problem is called a task, which can be an SOP or a MOP. Assuming that K minimization tasks are optimized simultaneously, multi-task optimization can be defined as Eq. (1).

$$\left\{\mathbf{x}_{1,}^{*}\mathbf{x}_{2,}^{*}...,\mathbf{x}_{K}^{*}\right\} = \left\{ argminT_{1}(\mathbf{x}_{1}), argminT_{2}(\mathbf{x}_{2}), ..., argminT_{K}(\mathbf{x}_{K}) \right\}$$
(2)

Where $T_j(j = 1, 2, ..., K)$ denotes the *j*th optimization task, \mathbf{x}_j represents the feasible solutions assigned to the *j*th task, \mathbf{x}_j^* indicates the optimal solution of the *j*th task.

Illuminated by the memetic computing and the multifactorial inheritance [38], Gupta et al. [15] proposed the fundamental EMT algorithm framework based on the memetic principle, namely the multifactorial evolutionary algorithm (MFEA), where each task $T_{je\{1,2,...,K\}}$ is considered as a memetic factor that influences the individual evolution in the *K*-factorial environment. To achieve efficient cross-domain exchange of genetic material between tasks, MFEA proposed the unified decision space strategy, which encodes each individual's decision variables into a unified space with an equal number of dimensions and uniform upper and lower boundaries to solve the isomerous decision space problem which each task has the diverse number of dimensions and each dimension has different boundaries. Specifically, during population initialization, each individual will be

encoded into the unified decision space with *D* dimensions, where *D* = max { $D_{je\{1,2,...,K\}}$ } is the largest number of dimensions of all the tasks, and each dimension is scaled to between [0,1]. By encoding into the unified decision space, one individual can be regarded as a combination of *K* chromosomes for different tasks. When solving a specific problem, the individual will be decoded into the corresponding target task space. Suppose \ddot{x}_i denotes the *i*th individual in the unified decision space, and x_i represents the result of \ddot{x}_i after decoding into the target task space. The decoding process is shown in Eq. (2).

$$\mathbf{x}_i = \ddot{\mathbf{x}}_i \left(1 : D_j\right) \odot \left(U_j - L_j\right) + L_j \tag{3}$$

Where $\ddot{x}_i(1:D_j)$ represents the first D_j dimensions of the *i*th individual in the unified decision space, and D_j is the number of dimensions of the *j*th task. U_j and L_j express the upper and lower bounds of the decision variables of the *j*th task, respectively. \odot denotes the Hadamard product.

To evaluate the individuals and compare the performance of individuals in different tasks, for each individual $p_{i \in \{1,2,...,|P|\}}$ in the population *P*, has the following definitions.

Definition 1. (*Factorial cost*): The factorial cost ψ_j^i is designed to measure the performance of indiv_idual p_i on the task T_j . When p_i is a feasible solution that satisfies the constraints of T_j , ψ_j^i is the value of the objective function of T_j . Otherwise, ψ_j^i is a very large real value, which means that p_i is not a feasible solution on task T_j and will be eliminated in the selection process.

Definition 2. (*Factorial Rank*): The factorial rank r_j^i expresses the fitness that p_i can solve the problem T_j . Specifically, r_j^i is the index of p_i after the population is arranged in the ascending order of ψ_i^i on task T_i .

Definition 3. (*Scalar Fitness*): The scalar fitness indicates the best performance that p_i can achieve in all the tasks and is calculated by the best factorial rank, as shown in Eq. (3).

$$\varphi_i = \frac{1}{\min_{j \in \{1, 2\dots, k\}} r_j^i} \tag{4}$$

Definition 4. *(Skill factor):* The skill factor τ_i represents the fittest task that p_i can solve, which is the index of the task that p_i achieves the best factorial rank, denoted as Eq. (4).

$$\tau_i = \operatorname{argmin}_j \left\{ r_j^i \right\} \tag{5}$$

Each individual in the unified decision space can be decoded to be dedicated to a specific task. When the population size is N, and the number of tasks is K, it is wasteful as the evaluation times will be N * K in one iteration because a solution cannot perform well on all the tasks. Therefore, an individual should ideally be evaluated only on the selected task most likely to perform well. Inspired by memetic computing [38], MFEA proposed the vertical cultural transmission mechanism to address this issue. The main idea is the offspring should share the same memetic environment with the parents by inheriting the task that the parents prefer. The offspring should have the same skill factor as the parent. This mechanism dramatically improves the efficiency of function evaluation. The frequency of function evaluation is reduced by K times compared with the case of evaluating all tasks for each individual.

MFEA provides the basic framework and theories for solving MTOPs, including unified decision space mechanism, solutions encoding and decoding method, and the concept of vertical cultural transmission. These concepts have a profound impact on the EMT algorithms proposed later. The proposed HMOMFMA is also based on the multifactorial framework.

2.2. Evolutionary multitasking

Since EMT is an emerging research field, researchers have worked on this from different aspects. To explore and exert the superiority of EMT, various tricks are applied to design and improve EMT algorithms. From the perspective of enhancing recombination strategies, Liu *et al.* [20] present a surrogate-assisted multi-tasking memetic algorithm that the surrogate model with Gaussian process to predict the optimal solution. Feng *et al.* [21] introduce the particle swarm algorithm and differential evolution (DE) algorithm into the EMT field. Song *et al.* [22] extend a dynamic multiswarm algorithm into the EMT area. Each task is arranged in an independent swarm, and then each swarm is partitioned into multiple sub-swarms, using swarm intelligence to optimize.

Considering identifying the most appropriate source task, Zhang *et al.* [23] introduce a probability model learned by the estimated distribution algorithm to represent the distribution of solutions and use the Wasserstein distance to evaluate the similarity between tasks. Chen et al. [24] introduce an archive strategy and a cumulative reward mechanism to measure task similarity by computing the kullback-leibler divergence of archives. Huang *et al.* [25] use the covariance matrix to characterize the distribution of historical solutions for a specific task and select the most suitable task through the similarity of the covariance matrix.

As for adjusting the frequency of knowledge transfer, Bali *et al.* [26] present a data-driven online learning method to optimize the transfer intensity. Zheng *et al.* [27] propose a novel concept named an ability vector to dynamically measure the correlation between the tasks to regulate the transfer intensity in the search process automatically. Li *et al.* [28] propose an adaptive transfer strength strategy, where the knowledge transfer strength is positively correlated with the transfer success rate.

Based on the view of properly allocating search resources, Gong *et al.* [29] propose a dynamic online resource allocation strategy based on problem difficulty, more complex problems will get more computing resources. Yao *et al.* [30] present an EMT algorithm based on decomposition and a dynamic resource allocation strategy, task with faster convergence will be allocated more computing resources. Wen *et al.* [31] propose that computing resources should be reallocated when knowledge transfer starts to fail.

Regarding search space mapping, Feng *et al.* [32] propose a novel task mapping mechanism based on denoising autoencoders (EMT-A), where the solution of the source task can be projected to the target task through the mapping matrix learned by the denoising autoencoder. Bali *et al.* [33] present a linearized domain adaptation method to transform the search space of simple tasks into a reconstruction space highly correlated with complex tasks. Ding *et al.* [34] introduce a novel decision variable transformation strategy to map the solutions of different tasks into a unified space.

As for multi-objective multitasking, the research is still in its infancy, and the literature is not abundant. Gupta et al. [35] first introduce the EMT theory into the multi-objective optimization area and propose the multi-objective multifactorial optimization algorithm (MOMFEA). The MOMFEA embeds the classic MOEA NSGA-II into the splendid EMT algorithm framework multifactorial evolutionary algorithm by analogizing the knowledge transfer as the transmission of cultural building blocks in memetic computation to optimize multiple MOPs simultaneously. Liang et al. [36] develop a novel multi-objective multifactorial algorithm based on subspace alignment and adaptive differential evolution named MOMFEA-SADE. In MOMFEA-SADE, a mapping matrix obtained by the subspace alignment strategy is introduced to transform the search space to reduce the probability of negative transfer, and an improved adaptive differential evolution is applied as the recombination operator to enhance search efficiency. Liang et al. [37] hybridize two novel strategies that genetic transform and hyper-rectangle search into an EMT algorithm, namely MFEA-GHS, which has proved excellent performance in multi-objective multitasking. The genetic



Fig. 1. The experiment illustrates the impact of information transfer using different strategies on the target task's decision variables and objective function values. (a) The decision variables values after the information transfer occurs on the convergence-related decision variables. (b) The objective values after the information transfer occurs on the convergence-related decision variables. (c) The decision variables values after the information transfer occurs on the diversity-related decision variables. (d) The objective values after the information transfer occurs on the diversity-related decision variables.

transformation strategy contributes to transforming search space by structuring mapping vectors to improve the efficiency of knowledge transfer. The hyper-rectangle approach constructed based on opposition learning is devoted to expanding the search ability in each subspace.

2.3. Motivation

In the MTOPs area, most of the current algorithms treat all the decision variables of the individual in the same way and randomly select the dimensions for information transfer while ignoring the characteristics of different decision variables and their impact on the individual. The decision variables in most MOPs can usually be classified into diversity-related decision variables and convergence-related decision variables. [46] Diversity-related decision variables are committed to the uniform distribution of solutions, and convergence-related decision variables are devoted to converging to the optimal point. In the MTOPs, if information transfer occurs between the convergence-related decision variables in the target task and the diversity-related decision variables in the source task, the distance between the decision variables of this dimension in the target population will increase, which is not conducive to the convergence of the target task. If information transfer occurs between the diversity-related decision variables in the target task and the convergence-related decision variables of the source task, the distance between decision variables of this dimension in the target population will shrink. This does not benefit the uniform distribution of the solution of the target task.

To illustrate the impact of the same and different types of decision variables on the efficiency of EMT during information transfer, an experiment is conducted on the CIHS problem in the classical MOMTO benchmark test suite [40], as shown in Fig. 1. The CIHS-T1 is adopted as the source task, and CIHS-T2 is adopted as the target task. Each task is a bi-objective optimization problem with 50 decision variables. In the algorithm's early and middle stages, the information transfer method is not activated to guarantee fairness. The last-generation population is applied with different transfer strategies.

In CIHS1 and CIHS2, the first decision variables are both diversityrelated, while the remaining decision variables are all convergencerelated. Fig. 1(a) shows the results of the information transfer of the convergence-related decision variables and diversity-related decision variables in the source task and the convergence-related decision variables of the target task. When the convergence-related decision variables transfer the information to the convergence-related decision variables, the target task still maintains remarkable convergence. But when the diversity-related decision variable in the source task shares information with the convergence-related decision variables in the target task, the convergence-related decision variables of the target task will diffuse to the surroundings. The direct consequence is that the solutions in the objective space will not converge to the true PF. Fig. 1(b) demonstrates the different performances of the solutions in the objective space conducted by the above two transfer strategies. Fig. 1(c) illustrates the results of the information transfer of the convergence-related decision variables and diversity-related decision variables in the source task and

Z. Xu et al.

the diversity-related decision variables in the target task. It can be seen that after the information transfer of the diversity-related decision variable in the target task and the convergence decision variables in the source task, the distances between individuals in the target task become smaller. The standard deviation of the diversity-related decision varinon-dominated solutions in each generation. And the proposed HMOMFMA also combines the EGS as the local search operator to improve the search capabilities. The overall framework of the proposed HMOMFMA is summarized in Algorithm 1.

Algorithm 1. The overall framework of HMOMFMA.

Input:	N: population size, N_D : non-dominated population size, N_A : active population size, N_C : clone								
populat	opulation size, K: number of tasks.								
Output: the final non-dominated solution set.									
1. Class	1. Classify the decision variables for each task.								
2. Initialize the population P.									
3. Assig	gn the skill factor for each individual in P evenly.								
4. Evalu	ate each individual in P according to the skill factor.								
5. Select all the non-dominated individuals to form the non-dominated population P_N									
6. Whi	e not reaching the maximum evaluation times do								
7.	Update the active population P_A .								
8.	Generate the clone population P_C .								
9.	Recombine individuals in P_C based on decision variable characteristics								
10.	Evaluate the offspring and add them into P_C '.								
11.	If $u < LS$ then, where u is a random number between [0,1].								
12.	r percent of individuals perform evolutionary gradient search.								
13.	Evaluate the offspring and add them to P_C '.								
14.	End if								
15.	Assign the skill factor to the individuals in P_C and evaluate.								
16.	Union P_N and P_C ', and select all the non-dominated individuals to update P_N .								
17. Enc	l while								

able in the target task is 3.113E-1 after shared information by the convergence-related decision variables, while the standard deviation is 3.124E-1 after shared information by the diversity-related decision variables. Fig. 1(d) demonstrates this point by contrasting the performance of the population in the objective space. The IGD value of solutions after taking in the information from the diversity-related decision variable is 2.471E-4, which is better than the solutions that receive the information from the convergence-related decision variables whose IGD value is 2.805E-4. Therefore, we can conclude that information transfer should be carried out with decision variables of the same type in the source task and the target task, which is beneficial to the effect of EMT. On the contrary, information transfer with decision variables of different types is likely to lead to negative transfer.

Therefore, if the algorithm can first classify the decision variables according to their characteristics related to diversity or convergence and then carry out the corresponding transfer method according to the category of the decision variable, the effect of information transfer can be significantly improved. This paper proposes an improved and efficient way to classify decision variables based on control variable analysis to divide decision variables into diversity and convergence two categories.

3. Proposed method

3.1. The overall framework of HMOMFMA

The characteristics of different decision variables should be utilized efficiently to improve the efficiency of information transfer. Based on this idea, in the proposed HMOMFMA, the decision variables of each task will be classified as diversity-related decision variables and convergence-related decision variables according to their contribution to the diversity and convergence of the population. Different evolutionary operators will be applied to the decision variable when generating new individuals according to their type. To accelerate the convergence and maintain the diversity, the proposed HMOMFMA hybridizes the immune algorithm [39], focusing on the sparsely distributed

First, unlike the classic EMT algorithm, HMOMFMA classifies the decision variables according to their contribution to convergence and diversity into two types. This strategy fundamentally determines the evolutionary operator applied to a specific decision variable, the detailed description of the classification method is presented in Section 3.2. Then the population is initialized, and each individual is encoded according to the unified decision space mechanism. All the individuals will be assigned to each task evenly. Each individual will be set the corresponding skill factor and be evaluated in its task. After that, the non-dominated solutions in each task will be picked out to form the nondominated population represented as P_N . In addition to ensuring convergence, diversity is crucial in the iteration process, so more attention should be paid to the sparse areas of the population. Based on this point, N_A solutions with the most sparse distribution in the objective space are selected from P_N to form the active population P_A as the parents of the clone population P_C . The sparse degree of the individual is evaluated by the maximum extension distance(MED), which has been proven to be an efficient and helpful method to assess the density of the solution, where a larger MED value indicates the solution is farther away from the other solutions in the objective space. The specific calculation method of MED is shown in Algorithm 2.

The proposed HMOMFEA follows the core idea of the immune algorithm named clonal selection mechanism, which is inspired by the massive asexual reproduction and mitosis of antibody cells in immunology [47]. In the immune system, the genes of the progeny cells are the same as those of the parent cells, which can enhance the binding to the antigen [48]. Clonal selection mechanism believes that better solutions should obtain more clone resources, and the purpose is to generate more local searches around the excellent solution [49]. The allocation of clone resources is carried out according to the MED of the individual in P_A , that the greater the MED value, the more clone resources it can gain. The mathematical model for clonal selection is shown in Eq. (5).

$$P_C = \bigcup_{i=1}^{N_A} \{h_i \bigotimes a_i\}, a_i \in P_A$$



Fig. 2. The frequency histograms of the values of 100 offspring generated by SBX and DE respectively.

(6)

$$h_i = \lceil N_C imes rac{MED(a_i)}{\sum\limits_{i=1}^{N_A} MED(a_i)}
ceil$$

where the operator \bigotimes indicates the cloning operator and the parameter

 h_i denotes the number of clones of each solution a_i in P_A . Next, the in-

the types of decision variables, as shown in Section 3.3. Afterward, following the rules of vertical cultural transmission in EMT, each individual in P_C' is assigned a skill factor and evaluated on the corresponding task. Finally, union the P_C' and P_N in the current generation and pick out all the non-dominated solutions to form the new P_N of the next generation.

Algorithm 2. The maximum extension distance.

Input: idx: the index of the current individual in the population, N: the population size, M: the number of objectives of the current task. Output: the MED of the current individual. 1. Set TotalDis=0. 2. Set NearDis=+ ∞ . 3. For *i*=1 to *N* 4. If (i==idx) then continue. 5. End if 6. Set Dis=0. 7. For *j*=1 to M $Dis=Dis+|f_i^{(i)}-f_i^{(idx)}|.$ 8 9. End for 10. TotalDis=TotalDis+Dis. If (Dis<NearDis) then NearDis = Dis. 11. 12. End if 13. End for 14. $MED = NearDis \times TotalDis$

dividuals in P_C will undergo recombination and local search to generate the offspring population P_C' . All the individuals in P_C will be recombined, and the *r* of the individuals in P_C will perform a local search when the local search threshold is reached. The local search adopts the EGS strategy, as described in Section 3.4. The recombination can be classified into intra-task recombination and inter-task information transfer, and different evolutionary operators are used for recombination according to

3.2. The decision variable classification method for information transfer

The specific pseudo code of proposed decision variable classification method is shown in Algorithm 3.

Algorithm 3. The proposed decision variable classification method.

```
Input: D: the number of the decision variables in a specific task, NS: the number of sampling
points from a single dimension of decision variables.
Output: IC: the set of indexes of the convergence-related decision variables, ID: the set of
indexes of the diversity-related decision variables.
1. IC=Ø, ID=Ø, S=Ø
2. For i=1 to D
         x_i = (u_i + l_i)/2.
3.
4. End for
5. For i=1 to D
         For j=1 to NS
6.
7.
                   \mathbf{X'} = \mathbf{X}.
                   x_i' = l_i + \frac{j}{N_s} * (u_i - l_i).
8.
                   add \mathbf{x}' to the archive S as \mathbf{s}_i and set n_i=0.
9
10.
         End for
11.
         Evaluate all the solutions in S, and set NF=0.
         For j=1 to NS
12.
13.
                   For k=1 to NS
14.
                            If (s_k \prec s_j) then
15.
                                      n_j = n_j + 1.
                            End if
16.
17.
                   End for
18.
                           = 0) then
                   If (n_i =
19.
                            NF=NF+1.
20.
                   End if
21
         End for
22.
         If (NF
                   = 1) then
23
                   add index i into IC.
24.
         Else
25
                   add index i into ID.
26.
         End if
27. End for
```

Firstly, a template solution for control variable analysis is initialized. All its dimensions are set to the mean value of the upper and lower bounds of the corresponding dimensions of the specific task, as shown in line 3. u_i and l_i represent the upper bound and the lower bound of the *i*th decision variable in this task respectively. Then, based on the template solution, NS solutions with different values of the ith variable but other dimensions remaining unchanged are generated to perform control variable analysis on the *i*th dimension, as shown in line 8, and these solutions are saved into the archive S. Next, the solutions in archive S are compared based on the dominance relationship. Suppose n_i , which represents the number of individuals dominating the *j*th individual in the archive, is zero. In that case, that means there not exists a solution that can dominate s_i in the archive S, and s_i is on the non-dominated front. And once such a non-dominated solution is found, NF plus one, where NF represents the total number of non-dominated solutions in archive S. Finally, if NF is equal to one, the ith dimension decision variable will be convergence-related, and the index i is stored in the IC set. Otherwise, the ith dimension decision variable is regarded as a diversity-related decision variable. The index *i* is stored in the *ID* set.

3.3. Recombination based on the decision variable characteristic

Effectively utilizing the characteristics of different decision variables can speed up the algorithm convergence and ensure the diversity of the population. The proposed HMOMFMA applies different recombination operators to the diversity-related and convergence-related decision variables. Regarding diversity-related decision variables, the offspring should be far away from the parent. For convergence-related decision variables, the offspring should be near the parent and performs a local search around the parent. Therefore, the differential evolutionary operator is the recombination operator for diversity decision variables. The DE operator will utilize the information of two additional solutions randomly selected from the population to optimize the current solution, and the generated offspring are less similar to the parent. For convergence decision variables, the SBX operator is applied to the recombination operator, and the generated offspring is near the parent, which can exploit more promising solutions in the local range while ensuring the effective convergence of the algorithm. Fig. 2 shows an example of the histogram of the offspring generated by the two operators respectively, where the values of the parent x_1 and x_2 are 0.3 and 0.7, and the value of the third parent x_3 in the DE operator is randomly selected from the decision space. The parameter η in SBX is set to 20, the parameter *F* in DE is set to 1, and both operators are tested 100 times respectively.

If the knowledge of other tasks can be efficiently utilized in the offspring generation process, the efficiency of the optimizing target task can be significantly improved. This is the main advantage and essential feature of the EMT algorithm. Thus, in the proposed HMOMFMA, the genetic mapping transfer strategy [37] is used to map individuals far apart in the unified decision space to the areas close to the target task. The genetic mapping transfer strategy can significantly improve the information transfer efficiency of the source task and reduce the negative transfer. The mathematical expression is shown in Eq. (6).

$$x_{i}^{target} = x_{j}^{source} \times \frac{\overline{x_{i}^{target} + \tilde{\epsilon}}}{\overline{x_{j}^{source} + \tilde{\epsilon}}}, i \in \{1, ..., D_{targer}\}, j \in \{1, ..., D_{source}\}$$
(7)

 x_i^{target} represents the transfer dimension in the target task, and $\overline{x_i^{target}}$

Table 1

Parameters setting for MOMTO experiments.

Parameter	Description	HMOMFMA	MOMFEA	MOMFEA-II	EMT-A	MFEA-SADE	MFEA-GHS	NSGA-II
rmp	Random mating probability	0.3	0.3	0.3	0.3	0.3	0.3	-
Ν	Population size for all the task	200	200	200	200	200	200	200
max FEs	Maximum number of evaluations	200,000	200,000	200,000	200,000	200,000	200,000	200,000
p_c	Crossover probability	0.9	0.9	0.9	0.9		0.9	0.9
η_c	Distribution index of crossover	20	20	20	20		20	20
p_m	Mutation probability	1/D	1/D	1/D	1/D		1/D	1/D
η_m	Distribution index of mutation	20	20	20	20		20	20
N_A	Population size of active population	100	-				-	-
N_C	Population size of clone population	200	-				-	-
F	Scaling factor in DE	0.5	-				-	-
ε	Step size factor in EGS	1.8	-				-	-
LS	Local search rate in EGS	0.02	-				-	-
r	Local search proportion in EGS	0.05	-				-	-
LST	Local search times in EGS	4	-				-	-
L	Number of trial solution in EGS	5	-				-	-

expresses the mean value of this dimension. x_j^{source} denotes the transfer dimension in the source task, $\overline{x_j^{source}}$ is the mean value of this dimension, and \tilde{e} indicates a tiny real value. Note that the transfer dimensions of the target task and the source task are not one-to-one correspondence but are selected according to the characteristics of the decision variables. When recombination proceedings, if the skill factors of parents are

completing the recombination, if the mutation condition is met, the polynomial mutation mutates the decision variable. The pseudo-code of the recombination method based on the decision variable characteristic is shown in Algorithm 4.

Algorithm 4. The recombination method based on the decision variable characteristic.

```
Input: a: parent individual from the P_C, IC: the set of indexes of the convergence-related decision
variables, ID: the set of indexes of the diversity-related decision variables.
Output: o: the generated offspring.
1. Randomly select two individuals b and c from the P_N.
2. For i=1 to D<sub>a</sub>,
         If i in ID
3.
4
                   Randomly select two diversity-related decision variables b_{div} and c_{div}.
5.
                   If (\tau_h != \tau_a) then
6.
                            Perform genetic mapping transfer strategy to b_{div}.
7.
                   End if
8.
                   If (\tau_c \mathrel{!=} \tau_a) then
9.
                            Perform genetic mapping transfer strategy to c_{div}.
10.
                   End if
                   o_i = DE(a_i, b_{div}, c_{div}).
11.
         End if
12.
13.
         If i in IC
14.
                   Randomly select a convergence-related decision variable b_{cov} from b.
15.
                   If (\tau_{\mathbf{b}} \mathrel{!=} \tau_{\mathbf{a}}) and rand1 < rmp then,
                            Perform genetic mapping transfer strategy to b_{cov}
16.
17.
                   Else o_i = polynomial mutation (a_i).
18.
                   End if
19.
                   o_i = SBX(a_i, b_{cov}).
         End if
20.
21.
         If rand2 < p_m then,
22.
                   o_i = polynomial mutation (o_i)
23.
         End if
24. End for
```

detected to be inconsistent, the genetic mapping transfer strategy will be activated to carry out the information transfer. One point that needs special attention is that since convergence-related decision variables require a local search in the vicinity of the parent, receiving all information from other tasks indiscriminately is inappropriate. The proposed HMOMFMA inherits the random mating probability (*rmp*) concept in MOMFEA. When *rmp* is met, cross-task knowledge transfer on convergence-related decision variables can be performed. If the skill factors of the parents are the same, the recombination is performed directly without activating the genetic mapping transfer strategy. After

3.4. Local search based on evolutionary gradient search

To effectively search for better solutions around high-quality solutions, which means the non-dominated solutions with sparse distribution, HMOMFMA uses a combination of global search and local search. The immune algorithm is applied as a global search optimizer, and EGS is used as a local search optimizer. The essential idea of the EGS is to follow the gradient information obtained in the evolution process to guide the population to move to optimal solutions. In SOPs, the deviation of the objective function value is usually used as the gradient information. In MOPs, combining the characteristics that the population commits to pressing close to the PF and evenly distributed, the proposed HMOMFMA applies the average value of the normalized objective function value as the fitness F(x) to evaluate the performance of the individual to obtain the gradient information in the iterative process. The specific pseudo code of EGS in HMOMFMA is shown in Algorithm 5.

Algorithm 5. Local search based on evolutionary gradient search.

information transfer is $O(dN_D)$. The time complexity of the cloning is $O(N_C)$, and the time complexity of the recombination and hypermutation is $O(dN_C)$. For the local search operator EGS, its time complexity is related to the depth of its search and the probability of occurrence. Its time complexity is $O(LS \times LST \times r \times L \times N_D)$. Overall, the total computational complexity of HMOMFMA is $(O(m(N_D + N_C)^2) + O(MN_A) + O(mN_D logN_D) + O(dN_D) + O(N_C) + O(dN_C)) \times O(LS \times LST \times r \times L \times N_D)$.

Input: a: the parent individual from the P_C , L: the number of trial solutions, LST: the total iteration times performing EGS. Output: ă: the final individual after local search. $1.\check{a}=a$ 2. For t=1 to LST For i=1 to L 3. Create trial solution \mathbf{r}_i by perturbing **ǎ** using mutation $N(0,\sigma_t^2)$. 4. 5 Set skill factor and evaluate r_i. 6. End for Assign fitness $F(\mathbf{r}_i)$ to each individual \mathbf{r}_i . 7. $\hat{v} = \frac{\sum_{l=1}^{L} [F(\mathbf{r}_l) - F(\check{\mathbf{a}})] \cdot (\mathbf{r}_l - \check{\mathbf{a}})}{||\sum_{l=1}^{L} [F(\mathbf{r}_l) - F(\check{\mathbf{a}})] \cdot (\mathbf{r}_l - \check{\mathbf{a}})||}, \text{ where } v \text{ is the estimated global gradient direction.}$ 8. 9 $\mathbf{o} = \check{\mathbf{a}} - \sigma_t \hat{\boldsymbol{v}}.$ $\sigma_{t+1} = \begin{cases} \sigma_t \varepsilon \\ \sigma_{t/\varepsilon} \end{cases}$ if $\mathbf{o} \prec \check{\mathbf{a}}$ 10. if $\check{\mathbf{a}} \prec \mathbf{o}$ 11 If $o \prec \check{a}$ then $\check{a} = o$ 12. End if 13. End for

EGS mainly consists of two steps, estimating the gradient direction through evolution and updating the solution using gradient descent. To estimate the gradient direction, first, the L trial solutions \mathbf{r}_i are generated by perturbing the parent by the normal distribution $N(0,\sigma_t^2)$, where σ_t controls the mutation strength. Then, according to the principle of vertical cultural transmission, the skill factor is assigned to \mathbf{r}_i and evaluated. The calculation of the gradient requires one single fitness index. Here the average of the normalized objective value is utilized as the fitness value. Afterward, the gradient is calculated according to the method shown in line 8. Next, the offspring solution o is generated using the gradient descent method, as shown in line 9. Then, according to the dominant relationship between the parent individual $\hat{\mathbf{a}}$ and the offspring individual **o**, the mutation step σ_t is updated. If the offspring **o** can dominate parent **a**, then the current **a** will be replaced by the offspring **o**, and the σ_t will be multiplied by the coefficient ε . Otherwise, the σ_t will be divided by the coefficient ε , where the coefficient ε is generally set to 1.8 [41]. Finally, when the local search times reach LST, the local search process is terminated, and the final individual a is output as the result of the local search.

3.5. Complexity analysis

In this section, the computational complexity of one generation of the proposed HMOMFMA is discussed. Suppose *d* denotes the total number of the dimensions of the unified decision space, N_D indicates the population size of the nondominated population, N_C expresses the population size of the clone population, N_A expresses the population size of the active population, *m* indicates the total number of the objective functions, *NS* is the number of sampling points from a single dimension of decision variables. Before the population is initialized, the decision variables will be classified. This classification is one-time, so the algorithm complexity is $O(dNS^2)$. In the environment selection, the time complexities of the nondominated sorting is $O(m(N_D + N_C)^2)$, the calculating of the MED is $O(mN_A)$, and the updating of the archive active population is $O(mN_D logN_D)$, respectively. The time complexity of the According to the operation rules of symbol *O*, the time complexity of the proposed HMOMFMA can be simplified as $O(m \times (N_D + N_C)^2 \times LS \times LST \times r \times L \times N_D)$. However, considering that the number and probability of local search usage is very small, in actual use, the speed of the algorithm will not be significantly reduced.

4. Experiments

4.1. Test suites introduction

In MTOPs, the similarity of the fitness landscape and the degree of intersection of optimal solutions are the two most important factors affecting the effectiveness of genetic information transfer between tasks. If the values of the corresponding dimensions of the optimal solutions of different tasks are closer, the transfer of genetic information between tasks is more conducive to optimization. Likewise, the more similar the fitness landscapes of optimization functions for different tasks are, the more knowledge individuals learn from source tasks can indirectly help optimize target tasks. According to the intersection degree of the global optimum, the classic MOMTO benchmark test problems are designed into three categories: complete intersection (CI), partial intersection (PI), and non-intersection (NI). According to the similarity of the fitness landscape, the classic MOMTO benchmark test problems can be divided into three categories of high similarity (HS), medium similarity (MS), and low similarity (LS). The classic MOMTO benchmark suite consists of nine consecutive multi-objective problems based on combining the above two classification strategies. Details of the classic MOMTO benchmark test suite can be found in [40]. The complex MOMTO benchmark test suite named CPLX is first introduced in the IEEE CEC 2019 competition on evolutionary multi-task optimization [42], which is more difficult than the classical test suite. The sub-problem of it is designed according to [43].

4.2. Compared algorithms

The proposed HMOMFMA will be compared with five state-of-the-art

multi-objective EMT algorithms MOMFEA [35], MOMFEA-II [26], EMT-A [32], MFEA-SADE [36], and MFEA-GHS [37] and a classic MOEA, namely NSGA-II [6]. MOMFEA [35] is the earliest and most classic multi-objective EMT algorithm, which can be regarded as the benchmark of multi-objective EMT algorithm. The well-known NSGA-II [6] is the basis for MOMFEA to apply the multitasking theory to process MTOPs. By comparing the performance of MOMFEA and NSGA-II, it can show the advantages of the multi-objective EMT algorithm over traditional MOEA, so NSGA-II is also included as one of the comparison algorithms. MOMFEA-II [26] applied a data-driven online learning method to optimize the transfer intensity during the search process to solve MOMTO problems and reduce negative transfers. EMT-A [32] applies the denoising autoencoder in the MFEA to map different tasks' decision spaces. MFEA-SADE [36] employs the subspace alignment strategy and adaptive differential evolutionary operator into MFEA. MFEA-GHS [37] is the improved version of MFEA that introduces genetic transform and hyper-rectangle search strategies into MFEA. All the algorithms are implemented utilizing the Jmetal 4.5.2 [45]. The platform on which the algorithm runs is a PC with Intel Core i5-9400 F CPU 2.90 GHz, and 16.00 GB of RAM.

4.3. Parameter settings

For a fair comparison, in MOMFEA, MOMFEA-II, EMT-A, MFEA-SADE, and MFEA-GHS, the population size is set to 200, but in NSGA-II, the population size for each task is set to 100. For the EMT algorithms, the maximum number of fitness evaluations is 200,000, but for the conventional MOEA like NSGA-II, the maximum number of evaluations for each task is set to 100,000. The parameter settings of the comparison algorithms are consistent with the original paper. The details of all the parameter settings are summarized in Table 1.

4.4. Performance indicators

For each benchmark problem, 10,000 points in the three-objective task and 1000 points in the two-objective task are sampling in true PF to evaluate the inverted generational distance (IGD) [44]. The IGD indicator measures the distance between the true PF and the closest individual in the obtained solutions. The indicator can be expressed as Eq. 7. where $Dist_i$ is the Euclidean Distance between the *ith* solution in the true PF and the closest individual in the obtained solutions and N_{PF} is the number of the sampling points. The lower the IGD value is, the better the convergence and diversity of the population.

Table 2

Averaged value and standard deviation of the IGD on the classical MOMTO test suite.

$$IGD = \frac{\left(\sum_{i=1}^{N_{PF}} Dist_i^2\right)^{1/2}}{N_{PF}}$$
(8)

4.5. Performance on classical MOMTO benchmark test suite

The experimental result of the classical MOMTO benchmark test suite is shown in Table 2. The average and standard deviation of IGD values over 20 independent runs of each algorithm are demonstrated, and the best result on each sub-problem is marked in gray. The Wilcoxon rank-sum test at the 95% confidence level was applied for the experimental results to compare the proposed HMOMFMA with other comparison algorithms. The significantly better, significantly worse, and not comparable results are represented using "+ ," "-," and "= ," respectively. In the classical MOMTO benchmark test suites, the multiobjective EMT algorithm can defeat the classic single-task MOEA NSGA-II on most benchmark test problems. This is mainly due to the knowledge-sharing and transferring mechanism of the EMT algorithm, which indicates that the multitasking optimization theory is indeed effective. From Table 2, compared to the state-of-the-art MOEAs MOMFEA, MOMFEA-II, EMT-A, MFEA-SADE, MFEA-GHS and NSGA-II, in terms of the IGD metric, the proposed HMOMFMA obtains superior results on 15, 15, 13, 15, 16 and 16 out of 20 sub-problems respectively in the complex MOMTO test suite.

Concerning the classical MOMTO benchmark test suite, the proposed HMOMFMA can perform best on high similarity (HS) problems such as CIHS-T1, PIHS, NIHS. This is because HMOMFMA classifies the decision variables according to their contribution to the diversity and convergence of the population and only transfers knowledge between decision variables of the same type between tasks, which makes the algorithm not interfere with the phased results of the target task and can effectively improve the knowledge reuse rate. When the two simultaneously optimized tasks possess high similarity, this strategy can significantly enhance algorithm performance. For problems with low similarities, such as PILS and NILS-T2, even if the coincidence degree of the optimum points is not high, the local search strategy based on EGS can constantly dig out potential better solutions to push the population forward. Therefore, HMOMFMA has also achieved significant advantages in these problems.

4.6. Performance on complex MOMTO benchmark test suite

The experimental results of the complex MOMTO benchmark test suite are shown in Table 3. The average and standard deviation of IGD

Problem	Task	HMOMFMA	MOMFEA	MOMFEA-II	EMT-A	MFEA-SADE	MFEA-GHS	NSGA-II
CIHS	T1	3.35E-04	3.80E-04(+)	4.69E-04(+)	3.65E-04(+)	3.89E-03(+)	1.57E-03(+)	1.97E-03(+)
	T2	8.64E-04	2.66E-03(+)	2.36E-03(+)	2.15E-04(-)	9.50E-04(+)	5.69E-03(+)	4.36E-03(+)
CIMS	T1	4.19E-02	6.62E-02(+)	1.76E-01(+)	1.82E-01(+)	3.26E-04(-)	1.09E-01(+)	1.14E-01(+)
	T2	5.72E-03	1.13E-02(+)	3.88E-04(-)	2.01E-02(+)	8.25E-04(-)	1.42E-02(+)	2.34E-02(+)
CILS	T1	8.30E-02	3.16E-04(-)	3.84E-04(-)	2.46E-04(-)	3.91E-03(-)	6.04E-01(+)	2.84E-01(+)
	T2	3.28E-04	1.98E-04(-)	3.95E-04(+)	1.90E-04(-)	5.01E-04(+)	4.16E-04(-)	2.02E-04(-)
PIHS	T1	4.38E-04	9.13E-04(+)	4.55E-04(+)	9.94E-04(+)	4.72E-04(+)	2.17E-03(+)	1.36E-03(+)
	T2	6.83E-03	2.51E-02(+)	1.48E-02(+)	3.95E-02(+)	2.22E-01(+)	5.89E-01(+)	5.91E-02(+)
PIMS	T1	1.86E-03	4.11E-03(+)	2.26E-03(+)	1.88E-03(+)	2.39E-02(+)	5.42E-03(+)	4.11E-03(+)
	T2	1.33E+01	1.53E+01(+)	1.36E+ 01(+)	2.52E+ 00(-)	1.52E+01(+)	1.55E+01(+)	1.60E+ 01(+)
PILS	T1	2.66E-04	3.18E-04(+)	4.57E-04(+)	3.07E-04(+)	1.10E-02(+)	5.25E-04(+)	2.75E-04(+)
	T2	5.56E-03	1.23E-02(+)	5.68E-03(+)	6.39E-02(+)	1.72E+ 00(+)	2.96E-02(+)	6.35E-01(+)
NIHS	T1	1.33E+00	1.56E+ 00(+)	1.52E+00(+)	1.57E+ 00(+)	4.46E+ 01(+)	5.38E+ 00(+)	5.21E+ 01(+)
	T2	2.42E-04	4.71E-04(+)	4.93E-04(+)	3.26E-04(+)	4.68E-04(+)	4.61E-03(+)	8.09E-04(+)
NIMS	T1	1.15E-01	4.18E-01(+)	3.76E-01(+)	1.88E-01(+)	1.41E+01(+)	2.81E-01(+)	4.40E-01(+)
	T2	6.31E-04	2.70E-02(+)	1.38E-02(+)	7.27E-04(+)	1.53E-03(+)	4.77E-02(+)	6.44E-02(+)
NILS	T1	1.07E-02	8.33E-04(-)	1.11E-03(-)	6.52E-04(-)	1.67E-02(+)	9.70E-04(-)	7.85E-04(-)
	T2	2.38E-04	6.43E-01(+)	6.42E-01(+)	4.88E-01(+)	1.92E+01(+)	6.43E-01(+)	6.42E-01(+)
			+ 15/- 3/= 0	+ 15/- 3/= 0	+ 13/- 5/= 0	+ 15/- 3/= 0	+ 16/- 2/= 0	+ 16/- 2/= 0

Table 3

Problem	Task	HMOMFMA	MOMFEA	MOMFEA-II	EMT-A	MFEA-SADE	MFEA-GHS	NSGA-II
CPLX1	T1	5.03E-04	4.31E-04(-)	1.87E-02(+)	2.54E-04(-)	2.86E-04(-)	5.37E-04(+)	5.26E-04(+)
	T2	8.82E-04	5.72E-03(+)	1.64E-01(+)	4.62E-03(+)	1.25E-03(+)	5.51E-03(+)	7.07E-03(+)
CPLX2	T1	3.54E-04	3.90E-04(+)	1.71E-02(+)	2.47E-04(-)	2.78E-04(-)	4.84E-04(+)	4.22E-04(+)
	T2	6.23E-03	4.15E-03(-)	1.70E-01(+)	1.27E-02(+)	6.81E-04(-)	1.22E-02(+)	1.04E-02(+)
CPLX3	T1	1.71E-03	5.41E-03(+)	1.57E-01(+)	3.68E-03(+)	1.88E-03(+)	4.54E-03(+)	4.94E-03(+)
	T2	1.81E-03	2.97E-03(+)	1.38E-01(+)	1.79E-03(-)	1.41E-03(-)	2.79E-03(+)	2.56E-03(+)
CPLX4	T1	1.26E-03	4.27E-03(+)	1.64E-01(+)	6.62E-03(+)	1.76E-03(+)	5.27E-03(+)	6.18E-03(+)
	T2	1.83E-03	4.73E-03(+)	1.95E-01(+)	8.03E-03(+)	2.17E-03(+)	4.50E-03(+)	6.73E-03(+)
CPLX5	T1	1.93E-03	2.32E-03(+)	1.07E-01(+)	5.19E-03(+)	2.19E-03(+)	2.68E-03(+)	2.54E-03(+)
	T2	3.20E-03	6.14E-03(+)	2.84E-01(+)	1.76E-02(+)	3.56E-03(+)	5.93E-03(+)	6.60E-03(+)
CPLX6	T1	1.15E-03	2.79E-03(+)	1.11E-01(+)	3.92E-03(+)	2.06E-03(+)	2.56E-03(+)	3.90E-03(+)
	T2	1.83E-03	5.18E-03(+)	2.01E-01(+)	5.46E-03(+)	3.31E-03(+)	4.94E-03(+)	7.11E-03(+)
CPLX7	T1	2.09E-03	2.54E-03(+)	1.47E-01(+)	2.21E-03(+)	1.37E-03(-)	2.57E-03(+)	2.58E-03(+)
	T2	9.02E-04	1.38E-03(+)	1.13E-01(+)	2.26E-03(+)	1.41E-03(+)	2.24E-03(+)	2.16E-03(+)
CPLX8	T1	8.75E-04	2.74E-03(+)	1.13E-01(+)	4.87E-03(+)	9.22E-04(+)	2.77E-03(+)	1.89E-03(+)
	T2	8.51E-03	1.12E-02(+)	3.55E-01(+)	7.62E-03(-)	2.24E-03(-)	8.58E-03(+)	9.97E-03(+)
CPLX9	T1	3.63E-03	6.17E-03(+)	2.94E-01(+)	3.21E-03(-)	4.25E-03(+)	7.35E-03(+)	6.26E-03(+)
	T2	1.21E-03	6.04E-03(+)	1.99E-01(+)	4.56E-03(+)	4.09E-03(+)	5.27E-03(+)	6.24E-03(+)
CPLX10	T1	9.56E-03	1.21E-02(+)	3.18E-01(+)	8.21E-03(-)	7.19E-03(-)	1.15E-02(+)	1.07E-02(+)
	T2	8.88E-03	9.70E-03(+)	2.31E-01(+)	9.67E-03(+)	9.69E-03(+)	9.25E-03(+)	1.08E-02(+)
			+ 18/- 2/= 0	+ 20 / - 0 / = 0	+ 14/- 6/= 0	+ 13/-7/= 0	+ 20 / - 0 / = 0	+ 20/-1/= 0

values over 20 independent runs of each algorithm are demonstrated, and the best result on each sub-problem is marked in gray. The Wilcoxon rank-sum test at the 95% confidence level was applied for the experimental results to compare the proposed HMOMFMA with other comparison algorithms. From Table 3, compared to the state-of-the-art EMT algorithms MOMFEA, MOMFEA-II, EMT-A, MFEA-SADE, and MFEA-GHS, in terms of the IGD metric, the proposed HMOMFMA obtains superior results on 18, 20, 14, 13, and 20 out of 20 sub-problems respectively in the complex MOMTO test suite. Compared to NSGA-II, the proposed HMOMFMA can get better results on all sub-problems.

The proposed HMOMFMA can achieve such good results on the complex MOMTO benchmark test suite because the knowledge transfer method based on decision variable classification can reduce the interference caused by the different types of decision variables transferring knowledge between tasks. The proposed HMOMFMA combines the immune algorithm and EGS as global and local search operators, respectively. The immune algorithm has strong convergence performance and maintains the diversity of the population through clone resource allocation. In contrast, EGS has good local search performance and can accelerate convergence by adapting the mutation length and guiding the

evolution according to the gradient information. The collaboration of global search and local search can significantly improve the convergence ability of the algorithm, making HMOMFMA perform best on the complex MOMTO benchmark test suite.

4.7. Discussion of proposed strategies

This section discusses the contribution of each proposed strategy to the performance of the proposed algorithm. Table 4 shows the results of decision variables classification on the classical and complex MOMTO benchmark test suites. Table 5 shows the IGD values of the multiobjective EMT evolution strategy algorithms with different operators on the classical MOMTO benchmark test suite, respectively. The average metric values over 20 independent runs of each algorithm are demonstrated, and the best result for each sub-problem is marked in gray. In addition, the Wilcoxon rank-sum test at the 95% confidence level was applied for the experimental results to compare the proposed HMOMFMA with other comparison algorithms. The significantly better, significantly worse, and not comparable results are represented using "+," "-," and "=," respectively. The MOMFMA only uses the multi-

Tuble 4	Tal	ble	4
---------	-----	-----	---

The results of decision variables classification on the mainstream benchm	arks
---	------

MTOP	Number of objectives	Number of variables	Diversity-related variables	Convergence-related variables
CIHS-T1, CIHS-T2,	2	50	<i>x</i> ₁	$x_2,, x_{50}$
CILS-T1, CILS-T2,				
PIHS-T1, PIHS-T2,				
PIMS-T1, PIMS-T2,				
PILS-T1, PILS-T2,				
NIHS-T1, NIHS-T2				
CIMS-T1	2	10	x_1	$x_2,, x_{10}$
CIMS-T2	2	10	$x_{1,} x_{9}$	$x_2, \ldots, x_{8}, x_{10}$
NIMS-T1	3	20	$x_{1,} x_{2}$	$x_3,, x_{20}$
NIMS-T2	2	20	$x_{1,} x_{2}$	$x_3,, x_{20}$
NILS-T1	3	25	$x_{1,} x_{2}$	x_3, \ldots, x_{24}
NILS-T2	2	50	$x_{1,} x_{2}$	$x_3,, x_{50}$
CPLX1-T1, CPLX2-T1	2	10	$x_{1,} x_{10}$	$x_2,, x_9$
CPLX1-T2, CPLX3-T1,	2	30	x_1	$x_2,, x_{30}$
CPLX3-T2, CPLX4-T1,				
CPLX4-T2, CPLX5-T1,				
CPLX6-T1, CPLX6-T2,				
CPLX7-T1, CPLX7-T2,				
CPLX8-T1, CPLX9-T2				
CPLX2-T2, CPLX8-T2,	2	10	<i>x</i> ₁₀	$x_1,, x_9$
CPLX10-T1, CPLX10-T2				
CPLX5-T2, CPLX9-T1	3	10	<i>x</i> ₁	$x_2,, x_{10}$

Table 5

Averaged	value of IGD	obtained by	MOMFMA,	MOMFMA-DVC,	MOMFMA-LS,	HMOMFMA	on the classi	cal MOMTO	benchmark test suite.
----------	--------------	-------------	---------	-------------	------------	---------	---------------	-----------	-----------------------

Problem	Task	MOMFMA	MOMFMA-DVC	MOMFMA-LS	HMOMFMA
CIHS	T1	1.11E-03(+)	9.89E-04(+)	4.20E-04(+)	3.35E-04
	T2	1.39E-03(+)	9.52E-04(+)	9.97E-04(+)	8.64E-04
CIMS	T1	1.05E-01(+)	1.00E-01(+)	8.94E-02(+)	4.19E-02
	T2	1.07E-02(+)	7.10E-03(+)	9.77E-03(+)	5.72E-03
CILS	T1	2.69E-01(+)	2.37E-01(+)	2.43E-01(+)	8.30E-02
	T2	3.93E-04(+)	3.66E-04(+)	3.56E-04(+)	3.28E-04
PIHS	T1	8.98E-04(+)	8.72E-04(+)	7.41E-04(+)	4.38E-04
	T2	2.95E-02(+)	2.67E-02(+)	1.08E-02(+)	6.83E-03
PIMS	T1	1.06E-02(+)	5.39E-03(+)	4.65E-03(+)	1.86E-03
	T2	4.25E+ 01(+)	4.14E+ 01(+)	3.78E+ 01(+)	1.33E+01
PILS	T1	2.87E-04(+)	2.75E-04(+)	2.69E-04(+)	2.66E-04
	T2	1.00E-02(+)	6.12E-03(+)	7.47E-03(+)	5.56E-03
NIHS	T1	5.54E+ 00(+)	4.78E+ 00(+)	2.26E+ 00(+)	1.33E+00
	T2	9.77E-04(+)	7.20E-04(+)	8.49E-04(+)	2.42E-04
NIMS	T1	2.74E-01(+)	1.31E-01(+)	1.25E-01(+)	1.15E-01
	T2	3.56E-03(+)	1.55E-03(+)	1.07E-03(+)	6.31E-04
NILS	T1	4.02E-02(+)	2.21E-02(+)	2.54E-02(+)	1.07E-02
	T2	8.84E-03(+)	4.73E-03(+)	4.60E-03(+)	2.38E-04
		+ 18/- 0/= 0	+ 18/- 0/= 0	+18/-0/=0	

Table 6	5
---------	---

Averaged value of IGD obtained by HMOMFMA with different LS on the classical MOMTO benchmark test suite.

Problem	Task	HMOMFMA $LS = 0.01$	HMOMFMA $LS = 0.02$	HMOMFMA $LS = 0.05$	HMOMFMA LS= 0.1	HMOMFMA LS= 0.2	HMOMFMA LS=0.4
CIHS	T1	4.28E-04(+)	3.35E-04	5.18E-04(+)	5.97E-04(+)	6.75E-04(+)	9.18E-04(+)
	T2	9.59E-04(+)	8.64E-04	1.02E-03(+)	1.06E-03(+)	1.40E-03(+)	5.13E-03(+)
CIMS	T1	1.53E-01(+)	4.19E-02	1.52E-01(+)	1.28E-01(+)	1.14E-01(+)	1.37E-01(+)
	T2	1.03E-02(+)	5.72E-03	6.49E-03(+)	1.29E-02(+)	1.39E-02(+)	8.48E-03(+)
CILS	T1	2.61E-01(+)	8.30E-02	3.25E-01(+)	2.39E-01(+)	2.55E-01(+)	4.45E-01(+)
	T2	3.75E-04(+)	3.28E-04	4.15E-04(+)	4.98E-04(+)	5.83E-04(+)	6.75E-04(+)
PIHS	T1	5.94E-04(+)	4.38E-04	8.07E-04(+)	9.78E-04(+)	3.83E-03(+)	2.95E-03(+)
	T2	1.18E-02(+)	6.83E-03	2.11E-02(+)	1.70E-02(+)	4.29E-02(+)	1.10E-01(+)
PIMS	T1	8.02E-03(+)	1.86E-03	8.74E-03(+)	7.85E-03(+)	1.47E-02(+)	1.44E-02(+)
	T2	3.55E+ 01(+)	1.33E+01	3.75E+ 01(+)	3.32E+01(+)	3.82E+ 01(+)	3.67E+ 01(+)
PILS	T1	2.95E-04(+)	2.66E-04	2.96E-04(+)	3.38E-04(+)	3.40E-04(+)	9.48E-04(+)
	T2	5.95E-03(+)	5.56E-03	6.40E-03(+)	7.79E-03(+)	1.05E-02(+)	9.59E-03(+)
NIHS	T1	3.43E+ 00(+)	1.33E+00	3.75E+ 00(+)	4.32E+ 00(+)	4.85E+ 00(+)	4.58E+ 00(+)
	T2	6.05E-04(+)	2.42E-04	5.86E-04(+)	9.02E-04(+)	1.84E-03(+)	2.70E-03(+)
NIMS	T1	1.29E-01(+)	1.15E-01	1.53E-01(+)	2.29E-01(+)	2.56E-01(+)	2.46E-01(+)
	T2	1.67E-03(+)	6.31E-04	3.07E-03(+)	2.04E-03(+)	1.79E-03(+)	6.77E-03(+)
NILS	T1	1.07E-02(=)	1.07E-02	1.07E-02(=)	1.06E-02(-)	1.07E-02(=)	1.08E-02(+)
	T2	2.54E-04(+)	2.38E-04	2.56E-04(+)	2.52E-04(+)	2.55E-04(+)	2.58E-04(+)
		+ 17/- 0/= 1		+ 17/- 0/= 1	+ 17/- 1/= 0	+ 17/- 0/= 1	+ 18 / - 0 / = 0

Table 7

Averaged	l value o	f IGD o	obtained	by	HMOMFMA	with	different	r on th	e classica	al MOMTO	benchmar	k test suite.
----------	-----------	---------	----------	----	---------	------	-----------	---------	------------	----------	----------	---------------

Problem	Task	HMOMFMA $r = 0.05$	HMOMFMA $r = 0.1$	HMOMFMA $r = 0.2$	HMOMFMA $r = 0.5$	HMOMFMA $r = 0.8$
CIHS	T1	3.35E-04	5.59E-04(+)	5.11E-04(+)	6.33E-04(+)	7.25E-04(+)
	T2	8.64E-04	9.98E-04(+)	9.52E-04(+)	9.53E-04(+)	1.39E-03(+)
CIMS	T1	4.19E-02	1.46E-01(+)	1.22E-01(+)	1.08E-01(+)	1.36E-01(+)
	T2	5.72E-03	1.03E-02(+)	8.61E-03(+)	7.75E-03(+)	7.54E-03(+)
CILS	T1	8.30E-02	2.69E-01(+)	3.02E-01(+)	2.98E-01(+)	4.28E-01(+)
	T2	3.28E-04	3.93E-04(+)	3.88E-04(+)	4.09E-04(+)	4.36E-04(+)
PIHS	T1	4.38E-04	5.98E-04(+)	7.91E-04(+)	7.08E-04(+)	8.63E-04(+)
	T2	6.83E-03	1.02E-02(+)	8.38E-03(+)	1.84E-02(+)	3.12E-02(+)
PIMS	T1	1.86E-03	7.68E-03(+)	7.72E-03(+)	9.04E-03(+)	1.01E-02(+)
	T2	1.33E+01	4.26E+ 01(+)	2.98E+ 01(+)	3.81E+ 01(+)	2.91E+ 01(+)
PILS	T1	2.66E-04	2.91E-04(+)	2.73E-04(+)	2.88E-04(+)	2.85E-04(+)
	T2	5.56E-03	6.71E-03(+)	6.12E-03(+)	7.47E-03(+)	1.00E-02(+)
NIHS	T1	1.33E+00	3.35E+ 00(+)	4.10E+ 00(+)	4.87E+ 00(+)	4.76E+ 00(+)
	T2	2.42E-04	6.62E-04(+)	7.14E-04(+)	9.10E-04(+)	1.06E-03(+)
NIMS	T1	1.15E-01	2.63E-01(+)	1.88E-01(+)	1.80E-01(+)	1.33E-01(+)
	T2	6.31E-04	1.31E-03(+)	1.18E-03(+)	1.22E-03(+)	1.55E-03(+)
NILS	T1	1.07E-02	1.07E-02(=)	1.07E-02(=)	1.07E-02(=)	1.05E-02(-)
	T2	2.38E-04	2.65E-04(+)	2.47E-04(+)	2.72E-04(+)	2.47E-04(+)
			+ 17/- 0/= 1	+ 17/- 0/= 1	+ 17/- 0/= 1	+ 17/- 1/= 0

objective multifactorial immune algorithm. MOMFMA-DVC applies the evolutionary recombination strategy based on decision variables' characteristics to MOMFMA. MOMFMA-LS hybrids the EGS as the local search operator to MOMFMA. The parameters of the comparison algorithm are consistent. It can be seen from Table 5 that the algorithm results of MOMFMA-DVC and MOMFMA-LS are better than MOMFMA in 18 sub-problems. This proves that the evolutionary recombination strategy based on decision variables' characteristics and EGS as a local search operator are valid. And the combination of the two strategies can greatly improve the performance of the algorithm.

4.8. Parameter sensitivity analysis

In order to ensure that the parameter settings used by the proposed algorithm are reasonable, in this section, the parameters *LS* and *r* involved in the local search operator of the algorithm are analyzed. A comparative experiment on the classical multi-objective multi-task optimization benchmark test suite is conducted. The average values over 20 independent runs of each algorithm are demonstrated, and the best result for each sub-problem is marked in gray. In addition, the Wilcoxon rank-sum test at the 95% confidence level was applied for the experimental results, where significantly better, significantly worse, and not comparable are indicated using "+ ", "-", and "= ", respectively. The IGD values of the classical MOMTO benchmark test suites are listed in Table 6 and Table 7, respectively.

5. Conclusion

This paper proposes a novel evolutionary multitasking algorithm for multi-objective optimization, namely HMOMFMA, by hybridizing the decision variable classification method, multi-objective immune algorithm, and evolutionary gradient search. The decision variable classification method is applied to classify the decision variables into convergence and diversity-related decision variables. Unique strategies will be used for different decision variables during recombination and information transfer. This can bring benefits for improving information transfer efficiency in multitasking optimization and accelerating convergence. The immune algorithm focuses on the most sparsely distributed nondominated solutions in the population, has strong convergence ability, and can guarantee the diversity of the population. The evolutionary gradient search method is introduced as the local search operator, which can accelerate convergence by adaptively adjusting the mutation length and guiding the evolution according to the gradient information. Comprehensive experiments are conducted on both the classical and complex MOMTO test suites. The proposed HMOMFMA is compared with five state-of-the-art multi-objective EMT algorithms, MOMFEA [35], MOMFEA-II [26], EMT-A [32], MFEA-SADE [36], and MFEA-GHS [37] and a classic MOEA, namely NSGA-II [6]. The experimental results demonstrate that the proposed HMOMFMA is superior to other advanced EMT algorithms.

However, some issues can still be considered in future work. First, the proposed HMOMFMA can extend to optimize three or more tasks simultaneously. The modules that can select the most appropriate task for transferring from multiple tasks should be designed. Second, the method of classifying decision variables can be improved, and dynamic classification can also be considered during the process of evolutionary iteration. Next, the proposed HMOMFMA mainly solves general MOPs. The improved HMOMFMA can also solve expensive MOPs, dynamic MOPs, large-scale MOPs, multimodal MOPs, and other more complex problems. Finally, the information transfer method in the proposed HMOMFMA can be further improved to reduce the negative transfer.

CRediT authorship contribution statement

xu zhiwei: Methodology, Project administration, Writing – original draft, Writing – review & editing. **Zhang Kai:** Methodology, Validation.

Xu Jiafeng: Data curation. Xu Xin: Methodology, Validation. Wu Ni: Software, Validation, Visualization. He Juanjuan: Methodology, Software.

Declaration of Competing Interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Data availability

Data will be made available on request.

Acknowledgments

The authors would like to thank Yuan Yuan, Yew-Soon Ong, Liang Feng, A.K. Qin, Abhishek Gupta, Bingshui Da, Qingfu Zhang, Kay Chen Tan, Yaochu Jin, and Hisao Ishibuchi to kindly provide the source code [40]. We would also like to thank Zhengping Liang, Jian Zhang, Feng Liang and Zexuan Zhu for providing the code of [37].

References

- K. Zhang, Z. Xu, G.G. Yen, L. Zhang, Two-stage multi-objective evolution strategy for constrained multi-objective optimization, 1–1, IEEE Trans. Evolut. Comput. (2022), https://doi.org/10.1109/TEVC.2022.3202723.
- [2] K. Qiao, K. Yu, B. Qu, J. Liang, H. Song, C. Yue, An evolutionary multitasking optimization framework for constrained multiobjective optimization problems, IEEE Trans. Evolut. Comput. 26 (2022) 263–277.
- [3] K. Qiao, K. Yu, B. Qu, J. Liang, H. Song, C. Yue, H. Lin, K.C. Tan, Dynamic auxiliary task-based evolutionary multitasking for constrained multi-objective optimization, IEEE Trans. Evolut. Comput. (2022) https://ieeexplore.ieee.org/abstract/ document/9774845/ (accessed November 17, 2023).
- [4] K. Qiao, J. Liang, Z. Liu, K. Yu, C. Yue, B. Qu, Evolutionary multitasking with global and local auxiliary tasks for constrained multi-objective optimization, IEEE/ CAA J. Autom. Sin. 10 (2023) 1951–1964.
- [5] K. Qiao, J. Liang, K. Yu, M. Wang, B. Qu, C. Yue, Y. Guo, A Self-Adaptive Evolutionary Multi-Task Based Constrained Multi-Objective Evolutionary Algorithm, IEEE Transactions on Emerging Topics in Computational Intelligence. (2023). https://ieeexplore.ieee.org/abstract/document/10032050/ (accessed November 17, 2023).
- [6] K. Deb, A. Pratap, S. Agarwal, T. Meyarivan, A fast and elitist multi-objective genetic algorithm: NSGA-II, IEEE Trans. Evolut. Comput. 6 (2002) 182–197, https://doi.org/10.1109/4235.996017.
- [7] K. Zhang, G.G. Yen, Z. He, Evolutionary algorithm for knee-based multiple criteria decision making, IEEE Trans. Cybern. 51 (2021) 722–735, https://doi.org/ 10.1109/TCYB.2019.2955573.
- [8] K. Zhang, Z. Xu, S. Xie, G.G. Yen, Evolution strategy-based many-objective evolutionary algorithm through vector equilibrium, IEEE Trans. Cybern. 51 (2021) 5455–5467, https://doi.org/10.1109/TCYB.2019.2960039.
- J. Bader, E. Zitzler, HypE: An algorithm for fast hypervolume-based many-objective optimization, Evolut. Comput. 19 (2011) 45–76, https://doi.org/10.1162/EVCO_a_ 00009.
- [10] W. Hong, K. Tang, A. Zhou, H. Ishibuchi, X. Yao, A scalable indicator-based evolutionary algorithm for large-scale multi-objective optimization, IEEE Trans. Evolut. Comput. 23 (2019) 525–537, https://doi.org/10.1109/ TEVC.2018.2881153.
- [11] N. Beume, B. Naujoks, M. Emmerich, SMS-EMOA: Multi-objective selection based on dominated hypervolume, Eur. J. Oper. Res. 181 (2007) 1653–1669, https://doi. org/10.1016/j.ejor.2006.08.008.
- [12] Hui Li Qingfu Zhang, MOEA/D: A multi-objective evolutionary algorithm based on decomposition, IEEE Trans. Evolut. Comput. 11 (2007) 712–731, https://doi.org/ 10.1109/TEVC.2007.892759.
- [13] M. Wu, K. Li, S. Kwong, Y. Zhou, Q. Zhang, Matching-Based selection with incomplete lists for decomposition multi-objective optimization, IEEE Trans. Evolut. Comput. 21 (2017) 554–568, https://doi.org/10.1109/ TEVC.2017.2656922.
- [14] Y. Yuan, H. Xu, B. Wang, B. Zhang, X. Yao, Balancing Convergence and diversity in decomposition-based many-objective optimizers, IEEE Trans. Evolut. Comput. 20 (2016) 180–198, https://doi.org/10.1109/TEVC.2015.2443001.
- [15] A. Gupta, Y. Ong, L. Feng, Multifactorial evolution: toward evolutionary multitasking, IEEE Trans. Evolut. Comput. 20 (2016) 343–357, https://doi.org/ 10.1109/TEVC.2015.2458037.
- [16] Lei Zhou, L. Feng, Jinghui Zhong, Y. Ong, Z. Zhu, E. Sha, Evolutionary multitasking in combinatorial search spaces: A case study in capacitated vehicle routing problem, IEEE Symp. . Ser. Comput. Intell. 2016 (2016) 1–8, https://doi.org/ 10.1109/SSCI.2016.7850039.

Z. Xu et al.

- [17] M.-Y. Cheng, A. Gupta, Y.-S. Ong, Z.-W. Ni, Coevolutionary multitasking for concurrent global optimization: with case studies in complex engineering design, Eng. Appl. Artif. Intell. 64 (2017) 13–24, https://doi.org/10.1016/j. engappai.2017.05.008.
- [18] R. Sagarna, Y. Ong, Concurrently searching branches in software tests generation through multitask evolution, IEEE Symp. . Ser. Comput. Intell. 2016 (2016) 1–8, https://doi.org/10.1109/SSCI.2016.7850040.
- [19] C. Wang, H. Ma, G. Chen, S. Hartmann, Evolutionary multitasking for semantic web service composition, IEEE Congr. Evolut. Comput. 2019 (2019) 2490–2497, https://doi.org/10.1109/CEC.2019.8790085.
- [20] D. Liu, S. Huang, J. Zhong, Surrogate-assisted multi-tasking memetic algorithm, IEEE Congr. Evolut. Comput. 2018 (2018) 1–8, https://doi.org/10.1109/ CEC.2018.8477830.
- [21] L. Feng, W. Zhou, L. Zhou, S.W. Jiang, J.H. Zhong, B.S. Da, Z.X. Zhu, Y. Wang, An empirical study of multifactorial PSO and multifactorial DE, IEEE Congr. Evolut. Comput. 2017 (2017) 921–928, https://doi.org/10.1109/CEC.2017.7969407.
- [22] H. Song, A.K. Qin, P.-W. Tsai, J.J. Liang, Multitasking Multi-Swarm Optimization, in: IEEE Congress on Evolutionary Computation, 2019, IEEE, Wellington, New Zealand, 2019, pp. 1937–1944, https://doi.org/10.1109/CEC.2019.8790009.
- [23] J. Zhang, W. Zhou, X. Chen, W. Yao, L. Cao, Multisource selective transfer framework in multiobjective optimization problems, IEEE Trans. Evolut. Comput. 24 (2020) 424–438, https://doi.org/10.1109/TEVC.2019.2926107.
- [24] Y. Chen, J. Zhong, L. Feng, J. Zhang, An adaptive archive-based evolutionary framework for many-task optimization, IEEE Trans. Emerg. Top. Comput. Intell. 4 (2020) 369–384, https://doi.org/10.1109/TETCI.2019.2916051.
- [25] S. Huang, J. Zhong, W.-J. Yu, Surrogate-assisted evolutionary framework with adaptive knowledge transfer for multi-task optimization, IEEE Trans. Emerg. Top. Comput. 9 (2021) 1930–1944, https://doi.org/10.1109/TETC.2019.2945775.
- [26] K.K. Bali, A. Gupta, Y.-S. Ong, P.S. Tan, Cognizant multitasking in multiobjective multifactorial evolution: MO-MFEA-II, IEEE Trans. Cybern. 51 (2021) 1784–1796, https://doi.org/10.1109/TCYB.2020.2981733.
- [27] X. Zheng, A.K. Qin, M. Gong, D. Zhou, Self-Regulated evolutionary multitask optimization, IEEE Trans. Evolut. Comput. 24 (2020) 16–28, https://doi.org/ 10.1109/TEVC.2019.2904696.
- [28] G. Li, Q. Zhang, W. Gao, Multipopulation Evolution Framework for Multifactorial Optimization. in: Proceedings of the Genetic and Evolutionary Computation Conference Companion, ACM, New York, NY, USA, 2018, pp. 215–216, https:// doi.org/10.1145/3205651.3205761.
- [29] M. Gong, Z. Tang, H. Li, J. Zhang, Evolutionary multitasking with dynamic resource allocating strategy, IEEE Trans. Evolut. Comput. 23 (2019) 858–869, https://doi.org/10.1109/TEVC.2019.2893614.
- [30] S. Yao, Z. Dong, X. Wang, L. Ren, A Multi-objective multifactorial optimization algorithm based on decomposition and dynamic resource allocation strategy, Inf. Sci. 511 (2020) 18–35, https://doi.org/10.1016/j.ins.2019.09.058.
- [31] Y. Wen, C. Ting, Parting ways and reallocating resources in evolutionary multitasking, IEEE Congr. Evolut. Comput. 2017 (2017) 2404–2411, https://doi. org/10.1109/CEC.2017.7969596.
- [32] L. Feng, L. Zhou, J. Zhong, A. Gupta, Y. Ong, K. Tan, A.K. Qin, Evolutionary multitasking via explicit autoencoding, IEEE Trans. Cybern. 49 (2019) 3457–3470, https://doi.org/10.1109/TCYB.2018.2845361.
- [33] K.K. Bali, A. Gupta, L. Feng, Y.S. Ong, Tan Puay Siew, Linearized domain adaptation in evolutionary multitasking, IEEE Congr. Evolut. Comput. 2017 (2017) 1295–1302, https://doi.org/10.1109/CEC.2017.7969454.

- [34] J. Ding, C. Yang, Y. Jin, T. Chai, Generalized multitasking for evolutionary optimization of expensive problems, IEEE Trans. Evolut. Comput. 23 (2019) 44–58, https://doi.org/10.1109/TEVC.2017.2785351.
- [35] A. Gupta, Y.S. Ong, L. Feng, K.C. Tan, Multi-objective multifactorial optimization in evolutionary multitasking, IEEE Trans. Cybern. 47 (2017) 1652–1665, https:// doi.org/10.1109/TCYB.2016.2554622.
- [36] Z. Liang, H. Dong, C. Liu, W. Liang, Z. Zhu, Evolutionary multitasking for multiobjective optimization with subspace alignment and adaptive differential evolution, IEEE Trans. Cybern. 52 (2022) 2096–2109, https://doi.org/10.1109/ TCYB.2020.2980888.
- [37] Z. Liang, J. Zhang, L. Feng, Z. Zhu, A hybrid of genetic transform and hyperrectangle search strategies for evolutionary multi-tasking, Expert Syst. Appl. 138 (2019) 112798, https://doi.org/10.1016/j.eswa.
- [38] X. Chen, Y.-S. Ong, M.-H. Lim, K.C. Tan, A multi-facet survey on memetic computation, IEEE Trans. Evolut. Comput. 15 (2011) 591–607, https://doi.org/ 10.1109/TEVC.2011.2132725.
- [39] M. Gong, L. Jiao, H. Du, L. Bo, Multi-objective immune algorithm with nondominated neighbor-based selection, Evolut. Comput. 16 (2008) 225–255, https://doi.org/10.1162/evco.2008.16.2.225.
- [40] Y. Yuan, Y.-S. Ong, L. Feng, A.K. Qin, A. Gupta, B. Da, Q. Zhang, K.C. Tan, Y. Jin, H. Ishibuchi, Evolutionary multitasking for multi-objective continuous optimization: benchmark problems, 1706.02766 [Cs], Perform. Metr. Baseline Results, ArXiv (2017). http://arxiv.org/abs/1706.02766.
- [41] C.K. Goh, Y.S. Ong, K.C. Tan, E.J. Teoh, An investigation on evolutionary gradient search for multi-objective optimization, in: IEEE Congress on Evolutionary Computation (IEEE World Congress on Computational Intelligence), IEEE, Hong Kong, China, 2008, pp. 3741–3746, https://doi.org/10.1109/CEC.2008.4631304, 2008.
- [42] L. Feng, K. Qin, A. Gupta, Y. Yuan, Y.-S. Ong, and X. Chi, IEEE CEC 2019 Competition on Evolutionary Multi-task Optimization, http://cec2019.org/ programs/competitions.html#cec-02 and http://www.bdsc.site/websites/MTO competiton 2019/MTO Competition CEC 2019.html, 2019.
- [43] Qingfu Zhang Hui Li, Multi-objective Optimization Problems With Complicated Pareto Sets, MOEA/D and NSGA-II, IEEE Trans. Evolut. Comput. 13 (2009) 284–302, https://doi.org/10.1109/TEVC.2008.925798.
- [44] P. Czyzżak, A. Jaszkiewicz, Pareto simulated annealing—a metaheuristic technique for multiple-objective combinatorial optimization, J. Multi-Criteria Decis. Anal. 7 (1998) 34–47, https://doi.org/10.1002/(SICI)1099-1360(199801)7:1.
- [45] J.J. Durillo, A.J. Nebro, jMetal: A Java framework for multi-objective optimization, Adv. Eng. Softw. 42 (2011) 760–771, https://doi.org/10.1016/j. advengsoft.2011.05.014.
- [46] X. Ma, F. Liu, Y. Qi, X. Wang, L. Li, L. Jiao, M. Yin, M. Gong, A Multi-objective evolutionary algorithm based on decision variable analyses for multi-objective optimization problems with large-scale variables, IEEE Trans. Evolut. Comput. 20 (2016) 275–298, https://doi.org/10.1109/TEVC.2015.2455812.
- [47] R. Shang, L. Jiao, F. Liu, W. Ma, A Novel immune clonal algorithm for MO problems, IEEE Trans. Evol. Comput. 16 (2012) 35–50, https://doi.org/10.1109/ TEVC.2010.2046328.
- [48] Y. Zhu, W. Li, T. Li, A hybrid Artificial Immune optimization for high-dimensional feature selection, Knowl. -Based Syst. 260 (2023) 110111, https://doi.org/ 10.1016/j.knosys.2022.110111.
- [49] X. Chen, J. Li, Y. Du, A hybrid evolutionary immune algorithm for fuzzy flexible job shop scheduling problem with variable processing speeds, Expert Syst. Appl. 233 (2023) 120891, https://doi.org/10.1016/j.eswa.2023.120891.